Author's response to reviews

Title: Patients presenting with somatic complaints in general practice: depression, anxiety and somatoform disorders are frequent and associated with psychosocial stressors

Authors:

Nader Haftgoli (nader.haftgoli-bakhtiari@chuv.ch)
Bernard Favrat (Bernard.Favrat@chuv.ch)
François Verdon (flverdon@net2000.ch)
Paul Vaucher (Paul.vaucher@gmail.com)
Thomas Bischoff (Thomas.Bischoff@hospvd.ch)
Bernard Burnand (Bernard.Burnand@chuv.ch)
Lilli Herzig (lilli.herzig@hin.ch)

Version: 8 Date: 14 September 2010

Author's response to reviews:

Dear Editor,

I made few corrections which do not impact the content of the manuscript:

- result section, line 9 : "ie" was replaced by "i.e."
- discussion section, 2nd paragraph, line 5 : "decreases" was replaced by "decreased"
- discussion section, 3rd paragraph, 3 lines before end : "to" was replaced by "than"

I also updated the acknowledgments section.

The manuscript is now ready for publication.

Regards

Nader Haftgoli